Author’s response to reviews

Title: Molecular Epidemiology and Hematologic Characterization of δβ-thalassemia and Hereditary Persistence of Fetal Hemoglobin in 125661 Families of Greater Guangzhou Area, the metropolis of Southern China

Authors:

FAN JIANG (496779026@qq.com)
Can Liao (canliaoj981@21cn.com)

Version: 7 Date: 17 Feb 2020

Author’s response to reviews:

Editor: Your manuscript "Molecular Epidemiology and Hematologic Characterization of δβ-thalassemia and Hereditary Persistence of Fetal Hemoglobin in 125661 Families of Greater Guangzhou Area, the metropolis of Southern China" (MGTC-D-19-00552R6) has been assessed. Many thanks for your response to our revision requests. However, there are a few final editorial requests we would like you to address before we accept the manuscript, detailed below.

1. Authors’ contributions

Please amend the initials “LJ” to “JL”.

Answer: We apologized for our carelessness. We have revised the manuscript according to the editor’s comment in line 3 of Authors’ contributions.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

Answer: We have revised the manuscript according to the editor's comment.